Author's response to reviews

Title: Analysis of the effectiveness of interventions used during the 2009 A/H1N1 influenza pandemic

Authors:

Nilimesh Halder (nilimesh@csse.uwa.edu.au)
Joel K Kelso (joel@csse.uwa.edu.au)
George J Milne (milne@cse.uwa.edu.au)

Version: 4 Date: 18 March 2010

Author's response to reviews:

Revisions made as per formatting changes request email of 17th March 2010:

1. Title in submission system updated to match manuscript (changed H1N1 to A/H1N1).

2. Reference 38 reordered (now number 33).

3. Erroneous reference to Figure 4 deleted, figure reference ordering is now correct.

4. Citation of tables in other references changed to eg "Table five of [35]" style instead of eg "Table 5 of [35]."

5. Full details of reference 30 have been added (this has now appeared in print so full details are included).

6. Vertical lines removed from table.

7. Vertically-merged table cells (covering multiple rows) have been removed.

8. Several minor typographic errors have been corrected.